Author's response to reviews

Title: Bone management in Japanese patients with prostate cancer: Hormonal therapy leads to increase the FRAX score

Authors:

Takashi Kawahara (takashi_tk2001@yahoo.co.jp)
Shusei Fusayasu (shu1299@yahoo.co.jp)
Koji Izumi (koji.izumi522@gmail.com)
Yumiko Yokomizo (yumiko1@yokohama-cu.ac.jp)
Hiroki Ito (pug_daikichi@yahoo.co.jp)
Yusuke Ito (t116008e@yokohama-cu.ac.jp)
Kayo Kurita (hirojiuemura@icloud.com)
Kazuhiro Furuya (kazuhiro.furuya1981@gmail.com)
Hisashi Hasumi (hisahasu@gmail.com)
Narihiko Hayashi (twnasru@yahoo.co.jp)
Yasuhide Myoshi (miyoyasu@yokohama-cu.ac.jp)
Hiroshi Miyamoto (hmiyamo1@jhmi.edu)
Masahiro Yao (masayao@yokohama-cu.ac.jp)
Hiroji Uemura (hu0428@yokohama-cu.ac.jp)

Version: 4
Date: 26 April 2016

Author’s response to reviews:

1. Please upload the figures as three individual figure files rather than all in one file

   Thank you for your suggestion. We upload figures in three separate files.

2. Remove the tables from the additional files and include them at the end of the main manuscript

   We appreciate your comment. We added tables in the main manuscript.

3. Please include an "Availability of data and materials" section after the Conclusions detailing where the data supporting the findings can be found. If you do not wish to share your data, please state in this section that data will not be shared, and state the reason.

   Thank you for your comment. We state the availability of data after conclusion section in the revised manuscript.

   In accordance with your suggestion, we deleted conclusion and incorporated it in our Discussion section.